

Citation: Fardousi N, Nunes da Silva E, Kovacs R, Borghi J, Barreto JOM, Kristensen SR, et al. (2022) Performance bonuses and the quality of primary health care delivered by family health teams in Brazil: A difference-in-differences analysis. PLoS Med 19(7): e1004033. https://doi.org/10.1371/ journal.pmed.1004033

Academic Editor: Margaret E. Kruk, Harvard University, UNITED STATES

Received: December 6, 2021

Accepted: May 26, 2022

Published: July 7, 2022

Peer Review History: PLOS recognizes the benefits of transparency in the peer review process; therefore, we enable the publication of all of the content of peer review and author responses alongside final, published articles. The editorial history of this article is available here: https://doi.org/10.1371/journal.pmed.1004033

Copyright: © 2022 Fardousi et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: The PMAQ scores for each family health team (our measure of quality of care) and the responses from a survey of RESEARCH ARTICLE

Performance bonuses and the quality of primary health care delivered by family health teams in Brazil: A difference-in-differences analysis

Nasser Fardousi¹, Everton Nunes da Silva², Roxanne Kovacs¹, Josephine Borghi¹, Jorge O. M. Barreto³, Søren Rud Kristensen⁴, Juliana Sampaio⁵, Helena Eri Shimizu², Luciano B. Gomes⁵, Letícia Xander Russo⁶, Garibaldi D. Gurgel⁷, Timothy Powell-Jackson¹*

1 Department of Global Health and Development, London School of Hygiene & Tropical Medicine, London, United Kingdom, 2 University of Brasília, Brasília, Brazil, 3 Oswaldo Cruz Foundation-Fiocruz, Brasília, Brazil, 4 Centre for Health Economics, University of Southern Denmark, Odense, Denmark, 5 Department of Health Promotion, Federal University of Paraiba, João Pessoa, Paraiba, Brazil, 6 Federal University of Grande Dourados, Mato Grosso do Sul, Brazil, 7 Oswaldo Cruz Foundation-Fiocruz, Pernambuco, Brazil

* timothy.powell-jackson@lshtm.ac.uk

Abstract

Background

Pay-for-performance (P4P) programmes to incentivise health providers to improve quality of care have been widely implemented globally. Despite intuitive appeal, evidence on the effectiveness of P4P is mixed, potentially due to differences in how schemes are designed. We exploited municipality variation in the design features of Brazil's National Programme for Improving Primary Care Access and Quality (PMAQ) to examine whether performance bonuses given to family health team workers were associated with changes in the quality of care and whether the size of bonus mattered.

Methods and findings

For this quasi-experimental study, we used a difference-in-differences approach combined with matching. We compared changes over time in the quality of care delivered by family health teams between (bonus) municipalities that chose to use some or all of the PMAQ money to provide performance-related bonuses to team workers with (nonbonus) municipalities that invested the funds using traditional input-based budgets. The primary outcome was the PMAQ score, a quality of care index on a scale of 0 to 100, based on several hundred indicators (ranging from 598 to 660) of health care delivery. We did one-to-one matching of bonus municipalities to nonbonus municipalities based on baseline demographic and economic characteristics. On the matched sample, we used ordinary least squares regression to estimate the association of any bonus and size of bonus with the prepost change over time (between November 2011 and October 2015) in the PMAQ score. We performed subgroup analyses with respect to the local area income of the family health team. The

municipality managers (our exposure variables) were provided to the authors through a collaborative agreement with the Department for Family Health at the Ministry of Health of Brazil. Requests for access to these data should be directed at the Department for Family Health: telephone number +55 61 33159044 or email desf@saude.gov.br. All other data used are publicly available at https://doi.org/10.17037/DATA. 00002886.

Funding: This research was funded by the Medical Research Council, Newton Fund and the Brazilian National Council for the States Funding Agencies (CONFAP) under the UK to Brazil Joint Health Systems Research Call (grant MR/R022828/1). The MRC grant was awarded to JB and TPJ. Funding from CONFAP came from Fundação de Amparo à Pesquisa do Distrito Federal (FAPDF), Fundação de Amparo à Ciência e Tecnologia do Estado de Pernambuco (FACEPE) and Fundação de Apoio à Pesquisa do Estado da Paraíba (FAPESQ). CONFAP funding was awarded to ES. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

Abbreviations: GDP, gross domestic product; OECD, Organisation for Economic Co-operation and Development; PMAQ, Programme for Improving Primary Care Access and Quality; P4P, pay-for-performance; STROBE, Strengthening the Reporting of Observational Studies in Epidemiology. matched analytical sample comprised 2,346 municipalities (1,173 nonbonus municipalities; 1,173 bonus municipalities), containing 10,275 family health teams that participated in PMAQ from the outset. Bonus municipalities were associated with a 4.6 (95% CI: 2.7 to 6.4; p < 0.001) percentage point increase in the PMAQ score compared with nonbonus municipalities. The association with quality of care increased with the size of bonus: the largest bonus group saw an improvement of 8.2 percentage points (95% CI: 6.2 to 10.2; p < 0.001) compared with the control. The subgroup analysis showed that the observed improvement in performance was most pronounced in the poorest two-fifths of localities. The limitations of the study include the potential for bias from unmeasured time-varying confounding and the fact that the PMAQ score has not been validated as a measure of quality of care.

Conclusions

Performance bonuses to family health team workers compared with traditional input-based budgets were associated with an improvement in the quality of care.

Author summary

Why was this study done?

- Pay-for-performance (P4P) programmes to incentivise health providers to improve quality of care have been widely implemented globally.
- P4P schemes vary considerably in how they are designed, but there is limited evidence on whether these design choices matter for quality of care.
- The National Programme for Improving Primary Care Access and Quality (PMAQ) in Brazil was a P4P scheme that gave municipalities autonomy to decide how funds could be spent and, specifically, whether payments could be used to reward health workers.

What did the researchers do and find?

- We used a difference-in-differences approach to examine whether performance bonuses given to family health team workers, compared with traditional input-based budgets, were associated with changes in the quality of care, and whether the size of bonus mattered.
- We found that giving bonuses to family health team workers was associated with an improvement in the quality of care, and the association increased with the size of bonus.
- Improvements in quality of care were most pronounced for family health teams located in the poorest two-fifths of areas.

What do these findings mean?

- The findings suggest that performance bonuses to family health team workers can potentially be a more effective way of using PMAQ funds to improve quality of care than input-based budgeting.
- Performance bonuses to family health team workers appeared to reduce inequalities in the delivery of primary health care.
- Further research is needed to better understand what other design features, such as who gets paid and the frequency of payment, influence the extent to which P4P schemes improve quality of care.

Please see S1 Portuguese Abstract for an alternate language Abstract.

Introduction

Primary health care is the foundation of many health systems. The vital role it plays, as a stepping stone towards achieving universal health coverage, is widely recognised [1]. Over the past two decades, Brazil has implemented sweeping primary health care reforms, of which the most high profile and consequential component was the Family Health Strategy [2,3]. According to this policy, family health teams spearhead primary health service provision at the community level free of charge [3]. Through public financing, family health teams were rapidly scaled up in communities across the country, resulting in improvements in population health [4–7]. Nonetheless, concerns over quality of care have persisted [8,9]. In 2011, Brazil introduced a national health financing programme to improve access to and quality of primary health care. Under this National Programme for Improving Primary Care Access and Quality (*Programa Nacional de Melhoria do Acesso e da Qualidade da Atenção Básica* [PMAQ]), the federal government made financial payments to municipalities based on the performance of family health teams.

Pay-for-performance (P4P) has been widely applied in the United States, the United Kingdom, other Organisation for Economic Co-operation and Development (OECD) countries, and increasingly in low- and middle-income countries [10–13]. The idea of linking financial payments to the performance of health providers has intuitive appeal. In practice, however, results from empirical studies are mixed, and key research questions remain unanswered, making it difficult to draw firm conclusions [14]. A possible explanation for the mixed results is that the specific design of P4P schemes can vary on many dimensions [15–17] and, although these design elements are likely of key importance for effectiveness, the evidence base for making informed choices is lacking [18].

PMAQ provides an ideal testing ground for addressing three questions about scheme design for which there is limited evidence. First, a key design decision in P4P schemes concerns how the money can be spent and, specifically, whether payments should be used to reward health workers. As a federal programme, many of the design features of PMAQ were national in scope. However, the programme was required to give municipalities discretion on how funds could be used, which means we can compare municipalities that gave performance bonuses to health workers with those that invested the funds entirely through traditional input-based budgets. Second, municipalities differed in the size of bonus paid to family health

team workers. Intuitively and theoretically, the size of incentive should matter for health service delivery and performance [19]. However, there are only a few studies in high-income countries that have examined the effect of bonus size [20,21]. It remains an open and pertinent question in low- and middle-income countries where health sector resources are more constrained [22,23]. Finally, because municipalities had the flexibility to decide whether to retain payments at the municipal level or redirect them to the family health team level, we can also speak to the question of whether varying the level of payment matters. Economic theory suggests that payments made closer to the executing level are more likely to be effective due to a reduced risk of free riding [24] and empirical evidence from one high-income country supports this hypothesis [25].

We examined whether performance bonuses given to family health team workers were associated with changes in the quality of care and whether the size of bonus mattered in Brazil. While PMAQ has been previously associated with reduced socioeconomic inequality in performance across teams, no study thus far has evaluated the impact of variations in PMAQ's design features across municipalities [26]. We hypothesised that performance bonuses to health workers provide a stronger incentive to improve the quality of primary health care than traditional input-based financing. Using national programme data, our primary analysis focused on family health teams providing care to approximately 35.4 million people.

Methods

Study setting and design

Family health teams are the lynchpin of the primary health care system in Brazil. They are the first point of contact for the community, providing primary health care to a catchment population of approximately 3,500 people. Each family health team operates from a health facility and comprises at least one physician, nurse, nurse assistant, full-time community health worker, and in some teams, dentist and oral health staff. PMAQ is described in more detail elsewhere [26]. In brief, it was a national programme that allocated around 10% of federal primary health care funds to municipalities based on the performance of family health teams [27]. It was implemented over three rounds between 2011 and 2019. At the beginning of each round, the performance of family health teams was assessed through a combination of self-assessment, routine monitoring, and independent external evaluation that was led by universities.

The assessment involved measurement of hundreds of indicators (598 in round 1 and 660 in round 3), some of which changed across the different rounds [28–30]. Indicators included those relating to service availability (e.g., opening hours), structural quality of care (e.g., availability of medicines), processes of care (e.g., content of care and treatment completion), outcomes (e.g., patient satisfaction and birth weight of children), utilisation of health care (e.g., patient volume), and management (e.g., appointments scheduled). Of the external evaluation indicators (n = 648) used in the third round of PMAQ, the most common were measures of structural quality (58.5%), followed by management practices (10.9%), clinical processes of care (10.7%), service availability (8.7%), outcome (8.3%), and utilisation (0.9%), with 2.0% unclassified.

Achievement of targets linked to each indicator was used to generate a summary measure of performance, known as the PMAQ score [28–30]. To calculate the score, the number of points achieved was divided by the number of points available in each of the three indicator categories. A weighted average across the categories was then multiplied by 100 to give the PMAQ score. The weights given to each indicator category changed between rounds, with slightly more weight given to routine monitoring indicators in round 3, at the expense of external evaluation indicators [26]. On the basis of this score, each participating family health team

was placed into a performance group that determined the monthly financial reward for the entire implementation round. The amount of money each municipality received was the sum of the specific rewards of family health teams within the municipality. A key feature of the federal design was that the performance groups were determined by the relative performance of family health teams within socioeconomic bands in the first two rounds of PMAQ. In round 3, performance groups were based simply on absolute PMAQ scores, with no adjustment for socioeconomic inequality.

Our study exploited the fact that municipalities, as the decentralised administrative health authority in Brazil, had autonomy in how PMAQ funds could be spent. Some municipalities chose to use some or all of the money to provide performance-related bonuses (henceforth bonus municipalities) to supplement the income of family health team members. In other words, financial incentives were passed down to the health provider level, with potential implications for worker motivation. In our sample, the majority of bonus municipalities (79.6%) stated that the bonuses were linked to performance on the external evaluation, and almost all (96.7%) gave the bonuses to every member of the family health team. The nonbonus municipalities spent the PMAQ funds in the traditional way using input-based budgets to purchase drugs and equipment and support infrastructure, training and management. This potentially improved facility readiness to provide care and the conditions of work but not the remuneration of health workers.

Our study design compared changes over time in quality of care between bonus and nonbonus municipalities. Specifically, we used a difference-in-differences approach combined with matching, a study design that has been shown to perform well in limiting bias [31] and has been widely applied in the evaluation of health policies. The matching sought to improve baseline balance between municipalities that passed on bonuses to family health team workers and those that did not give bonuses. Matching on pretreatment outcomes is attractive as it can improve balance for unobserved time-varying confounders [32,33]. The difference-in-differences method then controls for unobserved but fixed omitted variables, relying on the assumption that the counterfactual trends in the treatment and control groups are the same. The matching procedure makes this assumption of parallel trends more plausible by ensuring that the outcomes in the treatment and control groups are similar in levels at baseline [34–36]. Another useful property of matching is that it reduces bias from the potential misspecification of the subsequent regression model [37].

The study received ethics approval from the University of Brasilia (Brasilia, Brazil; CAAE 30424620.4.0000.8093), and the London School of Hygiene & Tropical Medicine (London, UK; 15805). The analysis in this study was planned in June 2018. A prospective study protocol or analysis plan is not available. This study is reported as per the Strengthening the Reporting of Observational Studies in Epidemiology (STROBE) guideline (S1 Checklist).

Data sources

We used data from five sources. First, we obtained the PMAQ score and performance category of all family health teams in each implementation round from the Ministry of Health. Second, to capture variation in the design of PMAQ, we used data from an online survey of municipality health managers, conducted as part of the external evaluation in the third round of implementation. This survey asked various questions on incentive design, including whether the municipality passed on PMAQ funds as bonuses to family health team workers and the size of the bonuses as a percentage of staff salaries. Third, we used the 2010 Brazilian Population Census to measure the average monthly income of households in each census area. We geographically linked each health facility to a census sector, allowing us to measure the local area income of each family health team [26]. Fourth, we obtained data on the characteristics of health facilities to which family health teams were attached from a census of health facilities done by the Ministry of Health in 2011. Fifth, we used established sources to construct a dataset of municipality socioeconomic and demographic characteristics for the year 2010 (S1 Table).

Measures

Our primary outcome was the PMAQ score, which we regard as a broad measure of quality of care. It was a composite measure based on indicators of service availability, structural quality of care, processes of care, outcomes, utilisation of healthcare, and management. The score, calculated by the Ministry of Health, could range from 0 (lowest possible score) to 100 (highest possible score) and was interpreted as the percentage of the maximum score obtainable by a family health team. The PMAQ score in round 3 was based on measurement of performance around October 2015. The PMAQ score in round 1 reflected performance around November 2011, at the beginning of the programme, before PMAQ funds had been disbursed. It therefore acted as a baseline.

The exposure variables were (i) whether the municipality used PMAQ funds to provide bonuses to family health team members; or (ii) categories indicating the size of bonus as a proportion of staff salaries (1% to 20%; 21% to 50%; more than 50%), with 0 % as the reference category. Some municipalities reported that the size of bonus they gave to teams varied and hence could not be categorised. These municipalities were dropped from the analysis of bonus size. Covariates included municipality characteristics (gross domestic product [GDP] per capita, human development index, Gini index, population, urban share of population, share of population under 5 years, share of population over 60 years, and average monthly PMAQ funds awarded per team in round 1), facility characteristics (type of health facility and number of clinical staff), and local area characteristics (monthly income per capita).

Statistical analyses

We used a difference-in-differences approach to examine the association between performance bonuses and quality of care. We analysed the data at the team level, creating a panel of teams that took part in the first and third round of PMAQ. Our regression models estimated the prepost change over time (the difference between round 3 and round 1) in the PMAQ score of family health teams in municipalities that gave bonuses relative to comparison municipalities that did not give bonuses. To explore whether there was a dose-response relationship, we replaced the binary exposure variable with dummy variables indicating the size of bonus. We fitted ordinary least squares regression. Robust standard errors were clustered at the municipality level given that exposure to bonuses varied at this level, as is standard in the health policy evaluation literature [33].

We controlled for the aforementioned covariates. One particular concern was that municipalities which gave bonuses may have also received more funding from PMAQ than those that did not [38]. By including the initial amount of PMAQ funding per team awarded to municipalities—an amount determined in the round 1 assessment at the beginning of the programme —we sought to deal with this potential source of confounding, thereby separating the incentive effect of bonuses from the influence of simply more financial resources. Because the models were estimated in first differences, the inclusion of the covariates meant we controlled for differential trends in quality based on initial values of these variables.

The key assumption underpinning any difference-in-differences approach is that the counterfactual outcomes for the treatment and comparison groups follow the same trend [39]. To increase the plausibility of the parallel trend assumption, we used propensity score methods to create a comparison group of nonbonus municipalities that best matched the bonus municipalities at baseline. We estimated propensity scores with probit regression using GDP per capita, human development index, Gini index, population, urban share of population, share of population under 5 years, share of population over 60 years, and the mean PMAQ score at baseline as predictors. We then performed one-to-one matching of municipalities, with no replacement and a calliper of 0.01. Bonus municipalities that could not be matched to a comparison municipality and nonbonus municipalities that were not the nearest match to a bonus municipality were discarded. To evaluate the matching procedure, we compared baseline balance between bonus and nonbonus municipalities and reported the *p*-value from a *t* test of the difference (see also \$2 and \$3 Tables and \$1 Fig). We report throughout results from a difference-in-differences approach without matching, in light of evidence suggesting that matching can introduce regression to the mean bias under certain restrictive conditions [35,40].

We conducted subgroup analyses to examine whether the association between bonuses and quality differed by the local area income of where family health teams were located. We categorised family health teams into five groups of equal size by local area income and included an interaction between this variable and treatment status (any bonus or size of bonus) in the main estimating equation. Using the margins command in Stata, we report the absolute effect for each subgroup, as well as the effect relative to the mean PMAQ score in round 1.

We performed several sensitivity analyses. First, the indicators and formula used to generate the PMAQ score varied across rounds. Although the difference-in-differences approach in principle deals with the change in measurement, the potential for bias remains. Based on the suggestion of a reviewer, we developed a structural quality of care index using a common set of 123 indicators from each PMAQ round that captured the availability of drugs, equipment, consumables, and diagnostic tests. We defined this index as the percentage of items available in each facility during the external assessment visit. We examined the sensitivity of our main findings to this alternative measure of performance. Second, we included additional control variables in the regression models. Third, we used a lagged dependent variable model as an alternative approach, since it is based on a different identifying assumption of conditional independence [32,39]. Fourth, we experimented with different callipers in the matching procedure. Fifth, we produced results for municipalities that gave bonuses to health workers but stated that the amount was not fixed. All analyses were done in Stata 16.1 SE.

Results

Of the 5,570 municipalities in Brazil, 5,028 (90.2%) implemented PMAQ and provided information on whether they gave bonuses to family health team members in round 3 (S2 Fig). We excluded 1,585 (31.5%) of 5,028 municipalities that had no family health team participating in both round 1 and round 3 of the programme and a further 72 (1.4%) municipalities that had no family health team with complete data on local area income and facility characteristics. Our unmatched analytical sample comprised 3,371 municipalities (1,937 nonbonus municipalities; 1,434 bonus municipalities), containing 13,716 family health teams (7,575 teams in nonbonus municipalities; 6,141 teams in bonus municipalities). After matching, the analytical sample comprised 2,346 municipalities (1,173 nonbonus municipalities; 1,173 bonus municipalities), containing 10,275 family health teams (5,052 teams in nonbonus municipalities; 5,223 teams in bonus municipalities). For the analysis of bonus size (S3 Fig), we excluded from the unmatched sample 610 municipalities whose size of bonus could not be categorised because it was reported to vary, leaving an unmatched analytical sample of 2,761 municipalities (1,937 with 0% bonus; 332 with 1% to 20% bonus; 359 with 21% to 50% bonus; 133 with >50% bonus) containing 11,060 family health teams (7,575 with 0% bonus; 1,197 with 1% to 20% bonus; 1,467 with 21% to 50% bonus; 821 with >50% bonus).

Table 1. Baseline characteristics of sample.

	Full (unmatched) sample			Matched sample			
	Any bonus to family health teams	No bonus to family health teams	p- Value	Any bonus to family health teams	No bonus to family health teams	p- Value	
Family health teams and local area							
Number of family health teams	6,141	7,575		5,223	5,052		
PMAQ score round 1	61.4 (9.3)	60.7 (10.4)	0.514	61.3 (9.5)	61.5 (10.1)	0.879	
Local area monthly income per capita	1.40 (0.82)	1.69 (0.88)	< 0.001	1.47 (0.84)	1.65 (0.95)	0.043	
Health facilities							
Number of facilities	5,523	5,988		4,626	3,859		
Facility type							
Health centre	4,189 (75.85%)	4,479 (74.8%)	0.480	3,590 (77.6%)	2,806 (72.7%)	0.008	
Health post and other	1,334 (24.2%)	1,509 (25.2%)		1,036 (22.4%)	1,053 (27.3%)		
Number of clinical staff	14.47 (6.83)	17.22 (11.46)	0.012	14.64 (7.1)	17.97 (12.78)	0.039	
Municipalities							
Number of municipalities	1,434	1,937		1,173	1,173		
PMAQ funds per FHT in round 1	4,849 (2141)	4,449 (2190)	< 0.001	4,718 (2,150)	4,639 (2,150)	0.375	
GDP per capita	10.86 (12.91)	14.13 (15.46)	< 0.001	11.99 (13.93)	11.89 (11.82)	0.840	
Human development index	0.65 (0.07)	0.68 (0.07)	< 0.001	0.66 (0.07)	0.65 (0.07)	0.546	
Gini index	0.51 (0.06)	0.49 (0.07)	< 0.001	0.50 (0.06)	0.51 (0.06)	0.484	
Total population	0.35 (1.04)	0.47 (3.21)	0.172	0.39 (1.14)	0.49 (3.92)	0.371	
Share of population urban	0.64 (0.22)	0.66 (0.22)	0.001	0.65 (0.22)	0.64 (0.21)	0.717	
Share of population under 5 years	0.07 (0.01)	0.07 (0.02)	< 0.001	0.07 (0.01)	0.07 (0.01)	0.869	
Share of population over 60 years	0.12 (0.03)	0.12 (0.03)	0.251	0.12 (0.03)	0.12 (0.03)	0.706	

Data are *n* (%) or mean (SD). PMAQ score is an index of quality between 0 and 100. Local area monthly income per capita is in Brazilian real divided by 1,000. Total population is divided by 100,000.

FHT, family health team; GDP, gross domestic product; PMAQ, National Programme for Improving Primary Care Access and Quality.

https://doi.org/10.1371/journal.pmed.1004033.t001

Table 1 presents descriptive statistics at baseline for bonus municipalities and nonbonus municipalities. In the full sample without matching, the mean PMAQ score was similar between treatment and control municipalities. However, the bonus municipalities received significantly more PMAQ funds per team and had lower GDP per capita and human development, greater income inequality, lower share of the population living in urban areas, and higher share of the population under the age of 5 years. By contrast, in the matched sample, there were no statistically significant differences between treatment and control for municipality characteristics, and the PMAQ score in round 1 was almost identical, indicating that the matching procedure achieved good balance.

Figs 1 and 2 show the mean PMAQ score in round 1 and round 3 in bonus and nonbonus municipalities as well as those categorised by size of bonus (see also S4 and S5 Figs). In the matched sample, the change over time in the mean PMAQ score was -1.65 points in control municipalities and 2.69 points in bonus municipalities, representing an unadjusted difference between the two groups of municipalities (difference-in-differences) of 4.3 points (95% CI 1.7 to 6.9; p = 0.001). The difference between control and the size of bonus groups of municipalities in the change over time in the mean PMAQ score was: 2.6 points (95% CI -1.6 to 6.7;





https://doi.org/10.1371/journal.pmed.1004033.g001

p = 0.231) for the small (1% to 20%) bonus size group; 6.7 points (95% CI 3.8 to 9.7; p < 0.001) for the medium (21% to 50%) size group; and 8.6 points (95% CI 5.5 to 11.7; p < 0.001) for the large (more than 50%) size group. The unadjusted difference-in-differences estimates on quality of care were similar, if a little larger, in the full sample without matching.

Table 2 presents the difference-in-differences regression estimates of the association between performance bonuses and quality of care. The results from the matched analysis show that the change over time in the PMAQ score was 4.6 points (95% CI: 2.7 to 6.4; p < 0.001) greater in the bonus municipalities compared with the nonbonus municipalities. This association was equivalent to a relative increase of 7.5% (over the baseline mean of 61.4 in the PMAQ score). The magnitude of the associations increased with the size of bonus, suggesting a dose–response relationship. The change over time in the PMAQ score was 8.2 points (95% CI: 6.2 to 10.2; p < 0.001) greater in the municipalities giving the largest bonuses compared with the nonbonus municipalities. The results from the analysis without matching were similar.

Fig 3 presents the subgroup effects with respect to local area income, revealing several patterns in the data (see also <u>S4 Table</u>). First, the association between bonuses and quality of care was U-shaped across the income distribution—that is, changes in quality of care were largest for teams in the poorest localities, fell as income of the catchment area increased, and then rose again in the richest quintile. Second, these differences were most pronounced when small bonuses were given, such that small bonuses were associated with better quality of care only for teams in the poorest two-fifths of areas. Third, large bonuses were associated with better



https://doi.org/10.1371/journal.pmed.1004033.g002

quality of care in teams across the income distribution and heterogeneity between income groups was less pronounced (S4 Table). Taken together, the results indicate that changes in quality were largest for teams in the poorest two-fifths of localities, but the size of bonus mattered most for teams in the richest three-fifths of localities.

We performed several sensitivity analyses. The pattern of results was similar when we used our structural quality of care index, although associations were smaller in magnitude (S5 Table). Results were also similar when we used a lagged dependent variable model in which we regressed the PMAQ score in round 3 on the incentive design indicator(s), baseline covariates, and the PMAQ score in round 1 (S6 Table). It is also worth noting that the coefficient on the initial amount of PMAQ funding per team awarded to municipalities was positive, implying a positive resource effect on quality. The results were generally not sensitive to the inclusion of additional controls, or the calliper value used in the matching. In some specifications, the difference-in-differences estimate for small (1% to 20%) bonuses was no longer significant but otherwise the findings were similar (S7 Table). We also report the results for municipalities who gave bonuses to health workers but stated that the amount was not fixed, with estimates slightly larger than the small (1% to 20%) bonus group.

Discussion

Our study examined the relationship between bonus payments to frontline primary health care workers and quality of care by exploiting variation in how municipalities decided to use

	Full (unmatched) sample				Matched sample			
	Any bonus to family health teams		Size of bonuses		Any bonus to family health teams		Size of bonuses	
	Coefficient (95% CI)	<i>p</i> -Value	Coefficient (95% CI)	<i>p</i> -Value	Coefficient (95% CI)	<i>p</i> -Value	Coefficient (95% CI)	<i>p</i> -Value
PMAQ bonus								
Municipalities giving bonuses	4.2 (2.6 to 5.7)	< 0.001			4.6 (2.7 to 6.4)	< 0.001		
PMAQ bonus size								
1 to 20% of salaries			2.6 (-0.4 to 5.6)	0.0841			3.1 (-0.1 to 6.4)	0.0609
21 to 50% of salaries			6.2 (4.3 to 8.1)	< 0.001			6.4 (4.1 to 8.6)	< 0.001
More than 50% of salaries			7.5 (5.6 to 9.3)	< 0.001			8.2 (6.2 to 10.2)	< 0.001
Local area								
Poorer	-1.5 (-2.4 to -0.7)	< 0.001	-1.4 (-2.3 to -0.5)	0.0029	-1.6 (-2.5 to -0.6)	0.0013	-1.5 (-2.6 to -0.5)	0.0039
Middle	-1.2 (-2.2 to -0.2)	0.0146	-1.1 (-2.2 to 0.0)	0.0580	-1.1 (-2.3 to 0.0)	0.0608	-0.8 (-2.1 to 0.5)	0.2061
Richer	-1.7 (-2.8 to -0.6)	0.0028	-1.8 (-2.9 to -0.6)	0.0028	-1.8 (-3.2 to -0.4)	0.0143	-1.9 (-3.4 to -0.5)	0.0091
Richest	-1.0 (-2.3 to 0.3)	0.1467	-1.4 (-2.6 to -0.2)	0.0226	-0.9 (-2.4 to 0.5)	0.2173	-1.5 (-2.8 to -0.2)	0.0275
Health facility								
Health centre	0.1 (-0.6 to 0.9)	0.7152	0.6 (-0.2 to 1.4)	0.1694	0.1 (-0.8 to 0.9)	0.8717	0.6 (-0.3 to 1.5)	0.2074
Number of clinical staff	-0.0 (-0.1 to 0.0)	0.3863	-0.0 (-0.1 to 0.0)	0.4906	0.0 (-0.0 to 0.0)	0.9160	0.0 (-0.0 to 0.0)	0.7443
Municipality characteristics								
PMAQ funds in round 1 (in R\$ 1,000)	-2.3 (-2.6 to -2.0)	< 0.001	-2.3 (-2.6 to -1.9)	< 0.001	-2.4 (-2.8 to -2.0)	< 0.001	-2.3 (-2.7 to -2.0)	< 0.001
GDP per capita (in R\$ 1,000)	-0.1 (-0.1 to -0.0)	0.0208	-0.0 (-0.1 to 0.0)	0.1108	-0.1 (-0.1 to 0.0)	0.1055	-0.0 (-0.1 to 0.1)	0.9148
Human development index	-13.7 (-28.8 to 1.4)	0.0758	-8.4 (-24.6 to 7.8)	0.3103	-14.7 (-31.8 to 2.5)	0.0936	-17.5 (-34.6 to -0.4)	0.0443
Gini index	-5.2 (-29.8 to 19.5)	0.6819	8.2 (-4.5 to 20.8)	0.2060	-12.9 (-44.6 to 18.8)	0.4249	3.5 (-12.4 to 19.4)	0.6668
Total population	0.0 (-0.0 to 0.1)	0.0964	0.0 (-0.0 to 0.0)	0.4417	0.0 (-0.0 to 0.1)	0.0671	0.0 (-0.0 to 0.1)	0.4060
Share of population urban	-2.9 (-6.3 to 0.6)	0.1009	-3.8 (-7.6 to -0.0)	0.0472	-2.1 (-6.3 to 2.1)	0.3260	-2.2 (-6.7 to 2.4)	0.3556
Share of population under 5 years	54.3 (-49.1 to 157.6)	0.3031	36.4 (-52.2 to 124.9)	0.4207	94.9 (-35.1 to 224.8)	0.1526	68.1 (-36.7 to 172.9)	0.2025
Share of population over 60 years	33.0 (-10.0 to 75.9)	0.1325	36.1 (-12.4 to 84.6)	0.1441	60.7 (10.0 to 111.4)	0.0190	72.5 (15.8 to 129.2)	0.0122
N teams	13,716		11,060		10,275		7,938	
N municipalities	3,371		2,761		2,346		1,836	
R-squared	0.171		0.186		0.177		0.200	

Table 2. Association between bonuses to family health team workers and the PMAQ score in the full (unmatched) and matched samples of municipalities.

The dependent variable is the change in the PMAQ score, which is an index of quality between 0 and 100. The reference groups are as follows: for PMAQ bonus is nonbonus municipalities; for PMAQ bonus size is nonbonus municipalities; for local area is poorest; and for health centre is health post and others. CI, confidence interval; FHT, family health team; GDP, gross domestic product; PMAQ, National Programme for Improving Primary Care Access and Quality.

https://doi.org/10.1371/journal.pmed.1004033.t002

funds under PMAQ. We found that giving bonuses to workers was associated with a significant increase in quality of care as measured by the PMAQ score. Improvements in quality of care were most pronounced for family health teams located in the poorest two-fifths of areas. The association with quality of care increased with size of bonus, suggesting a dose–response relationship. It is important to emphasise that we did not evaluate PMAQ, and the results should not be interpreted as estimates of the impact of the programme.

Compared with the control group, the PMAQ score in bonus municipalities increased by 4.6 points, equivalent to a relative increase of 7.5%. It is difficult to directly compare findings across studies because of the wide range of quality of care outcomes used in the literature. Our study connects most closely to the P4P literature in which studies have compared P4P with equivalent levels of input-based funding to disentangle the resource effect from the incentive

	Effect size	Relative to base
Any bonus to family health teams		
Poorest	 ← 6.8 (5.3, 8.3) → 7.4 (5.6, 9.2) 	11.8% 12.8%
Poorer	4.8 (2.7, 7.0) 5.4 (2.9, 8.0)	8.1% 9.1%
Middle	2.9 (0.7, 5.2) 2.7 (0.3, 5.0)	4.8% 4.3%
Richer	1.9 (-0.3, 4.0) 2.2 (-0.4, 4.8)	3.0% 3.5%
Richest	4.0 (1.7, 6.3) 4.6 (2.0, 7.2)	6.3% 7.1%
Bonus size: 1 to 20% of salaries		10010100
Poorest	6.5 (4.4, 8.6) 7.8 (5.3, 10.3)	11.3% 13.5%
Poorer	3.8 (1.5, 6.0) 5.3 (2.7, 7.9)	6.3% 8.9%
Middle	-0.8 (-6.0, 4.3) -0.5 (-5.1, 4.1)	-1.4% -0.8%
Richer	0.2 (-5.3, 5.7) 0.2 (-6.0, 6.4)	0.3% 0.3%
Richest	- 1.7 (-2.6, 6.0) 2.5 (-1.8, 6.9)	2.6% 3.9%
Bonus size: 21 to 50% of salaries		
Poorest	8.0 (6.1, 9.9) 8.9 (6.7, 11.1)	13.9% 15.5%
Poorer	7.2 (5.0, 9.4) 7.0 (4.3, 9.7)	12.1% 11.7%
Middle	5.5 (1.5, 9.5) 4.8 (0.3, 9.3)	9.1% 7.8%
Richer	2.6 (-1.0, 6.2) 2.6 (-1.4, 6.7)	4.2% 4.1%
Richest	6.7 (3.8, 9.6) 7.3 (4.1, 10.5)	10.6% 11.3%
Bonus size: more than 50% of salaries		
Poorest	8.9 (6.3, 11.5) 10.0 (6.8, 13.2	15.5% 2) 17.4%
Poorer	7.4 (4.8, 9.9) 8.1 (5.3, 10.9)	12.4% 13.6%
Middle	7.8 (4.8, 10.8) 7.6 (4.5, 10.7)	12.9% 12.3%
Richer +	5.2 (3.0, 7.4) 6.1 (3.7, 8.6)	8.4% 9.7%
Richest	8.6 (6.1, 11.1) 9.2 (6.6, 11.9)	13.5% <u>14.4</u> %
-10 0	1 10	

Fig 3. Income subgroup analyses. Red bars represent point estimates and 95% confidence intervals in the unmatched sample; blue bars represent point estimates and 95% confidence intervals in the matched sample. Income subgroups were defined using the average monthly income of households in the local area of a family health team. CI, confidence interval.

https://doi.org/10.1371/journal.pmed.1004033.g003

effect. In Rwanda, P4P increased tetanus vaccine during antenatal care by 5.1 percentage points, a composite antenatal content of care index by 0.16 standard deviations, and HIV testing by 10.4 percentage points [38,41]. Contrary to our study, an equity analysis found that impacts were greatest among the richest, although these results pertain to utilisation rather than quality of care outcomes [42]. In Zambia, P4P was found to have no significant effect on

the availability of inputs (facility infrastructure, drugs, or equipment), process quality of care (antenatal and child health care exit interviews), and client satisfaction [43]. In Cameroon, there was no evidence of an effect of P4P on process quality of care measured using direct observation of antenatal and childcare consultations [44]. Finally, in Benin, P4P had a significant effect on various aspects of clinical care, including a 5.5 percentage point, 2.7 percentage point, and 8.8 percentage point increase in checklists for history taking, physical examinations and advice during antenatal care consultations [45].

Our findings on bonus size connect to a small literature from the US. A study on the Medicare Advantage Quality Bonus Payment Demonstration programme found that doubling the size of payment did not result in better quality of care, possibly because incentives to providers were passed through insurers [20]. A small cohort study found that increasing bonus size by a mean of \$3,355 per doctor improved clinical quality of care by 3.2 percentage points [21]. Our study further contributes to this literature by examining changes in quality of care by size of bonus and wealth group, showing that the size of the bonus mattered most for family health teams located in richer areas, but improvement in quality of care in poorer areas was achieved with relatively small bonus payments.

How may bonuses to family health teams have improved quality of care? There are several plausible explanations. First, the bonuses may have increased the motivation of health workers and other family health team members, resulting in greater effort and less shirking. The indicators within the PMAQ score most amenable to worker effort are those concerned with service availability, processes of care, management practices, and possibly utilisation of health care. Second, the channel of influence may have been through family health teams exercising pressure on the health management at the municipality level to improve availability of equipment and medicines. The results from the analysis of the structural quality of care index provide evidence for this second channel. The smaller coefficients, however, also suggest that bonuses may have been more influential in improving indicators of service availability, processes of care, and management practices. Our subgroup results with respect to local area income may be explained by the fact that poorer areas had greater room for improvement (reflecting substantial socioeconomic inequalities in the Brazilian health system) [26,46], hence more potential for the bonuses to bite.

The strengths of the study are the national scale of the analysis, the use of longitudinal data on family health team performance for our quasi-experimental approach, and the availability of a fine-grained measure of local area income with which to examine subgroup effects. Our study has several limitations. First, our study design cannot rule out unmeasured confounding and is therefore unable to provide definitive evidence of a causal relationship between bonuses and improvement in quality of care. It remains possible that the improvement in quality of care in bonus municipalities relative to nonbonus municipalities was due to other time-varying factors that differentially affected the two sets of municipalities. Candidates include political factors, working arrangements, local contracting, existing salaries and other bonuses. Our analysis did, however, control for time-invariant factors at the team level alongside a rich set of municipality characteristics and achieved good baseline balance through the matching procedure. Evidence of a dose–response relationship with respect to bonus size provides further support for a causal interpretation of the findings.

Second, the PMAQ score is not a validated measure of quality. We do not know whether it is a predictor of health outcomes, nor can we be sure it is not vulnerable to gaming on the part of health providers. Whenever measures of performance are linked to financial reward, there is an incentive for gaming. We used the PMAQ score because it is the Brazilian Government's official measure of performance, was developed through a deliberate process with wide consultation, and was based largely on indicators collected independently by universities. Third, our measure of exposure was based on self-reported data on scheme design collected through an online questionnaire at one point in time. Not only could municipalities have changed how they used PMAQ funds during the study period, recall bias and other sources of measurement error may have affected the reliability of responses, particularly those regarding bonus size. Our large sample size and use of broad bonus size categories will have helped address these issues. Fourth, while we had data on whether bonuses were given by municipalities, we lacked information on how the funds were otherwise used, making a more nuanced interpretation of the findings challenging.

The findings of this study have several implications for policy makers. Because we compared how municipalities used PMAQ funds, while controlling for any differences in the amount of money received, there is no obvious need to assess the cost-effectiveness of bonus payments in the context of our study. Our findings imply that bonuses to family health team workers may be a more effective way of using PMAQ funds than more traditional input-based approaches. However, the important caveat is that we do know whether improvements in the PMAQ score translate into better health and patient experience. Moreover, policymakers must also consider whether the apparent benefit of performance bonuses is likely to fade over time and what the consequences would be for motivation if the bonuses were ever to be withdrawn. Another implication concerns health inequalities in Brazil, which present major health care challenges and were the driver behind introducing PMAQ. While PMAQ was previously found to reduce social inequalities [26], the findings of this study suggest that bonuses may have been a contributor to the overall PMAQ's redistributive effect. It is still worth noting that the short-term gains from P4P can decay in the long run [47]. The Brazilian government has recently rolled out a new primary health care financing scheme, Previne Brasil, to replace PMAQ. It has, however, retained P4P as a central element of the financing mechanism. Findings of this study can help inform design choices going forward.

To conclude, our findings show that giving performance bonuses to staff compared with traditional input-based budgets can potentially lead to a greater improvement in quality of care. This study provides an important contribution to the literature on P4P design with implications for policy makers. Given the wide range of features of P4P design, future research should focus on the effect of other P4P design features, such as payment frequency and recipients as either individual determinants of quality or in combination.

Supporting information

S1 Checklist. STROBE checklist completed with section and paragraph numbers for each item. STROBE, Strengthening the Reporting of Observational Studies in Epidemiology. (DOCX)

S1 Portuguese Abstract. Abstract in Portuguese. (DOCX)

S1 Table. Sources of data and variable descriptions. (DOCX)

S2 Table. Probit model used to calculate propensity scores. The probit regression was run on municipality level data. CI, confidence interval; GDP, gross domestic product; PMAQ, National Programme for Improving Primary Care Access and Quality. (DOCX)

S3 Table. Standardised bias before and after matching. The standardised % bias is the % difference of the sample means in the treated and nontreated (full or matched) subsamples as a

percentage of the square root of the average of the sample variances in the treated and non-treated groups.

(DOCX)

S4 Table. Income subgroup analysis differences. The table presents the income subgroup effects as the difference between subgroups (with the poorest group acting as the reference category). Rather than reporting the *p*-value on each subgroup effect, we report the *p*-value from a Wald test that these income subgroup coefficients are jointly equal to zero. CI, confidence interval; PMAQ, National Programme for Improving Primary Care Access and Quality. (DOCX)

S5 Table. Difference-in-differences results for structural quality of care. The dependent variable is the change in the structural quality of care score, which is an index of quality between 0 and 100. The reference groups are as follows: for PMAQ bonus is nonbonus municipalities; for PMAQ bonus size is nonbonus municipalities; for local area is poorest; and for health centre is health post and others. CI, confidence interval; FHT, family health team; GDP, gross domestic product; PMAQ, National Programme for Improving Primary Care Access and Quality.

(DOCX)

S6 Table. Lagged dependent variable results. Results are from a lagged dependent variable model based on the full, unmatched, panel of family health teams. The dependent variable is the PMAQ score in round 3. Regressions are at the level of family health teams, with standard errors clustered at the municipality level. The reference groups are as follows: for PMAQ bonus is nonbonus municipalities; for PMAQ bonus size is nonbonus municipalities; for local area is poorest; and for health centre is health post and others. CI, confidence interval; FHT, family health team; GDP, gross domestic product; PMAQ, National Programme for Improving Primary Care Access and Quality. (DOCX)

S7 Table. Other robustness checks. Each panel is a single robustness check, reporting results for the "any bonus" analysis and results for the "size of bonus" analysis. Panel A reports the results from the main analysis in the paper. Panel B is based on the unmatched sample and includes as additional controls: health care spending per capita and whether the political party of the municipality is the same as the national government. Panel C includes additional controls but is based on the matched sample. Panel D uses a smaller calliper of 0.001 in the matching procedure. Panel E uses a larger calliper of 0.2 in the matching procedure. Panel F reports results for size of bonus, including municipalities that gave a variable bonus amount to family health teams. CI, confidence interval. (DOCX)

S1 Fig. Histogram of propensity score of treated (bonus) and untreated (nonbonus) municipalities.

(TIF)

S2 Fig. Study flow diagram: any bonus. In the third step, some family health teams had no data on local area income because of missing geographical information to link them to the census area.

(TIF)

S3 Fig. Study flow diagram: size of bonus. In the fourth step, some family health teams had no data on local area income because of missing geographical information to link them to the

census area. (TIF)

S4 Fig. Violin plot of the PMAQ score by bonus status in the matched sample. PMAQ, National Programme for Improving Primary Care Access and Quality. (TIF)

S5 Fig. Violin plot of the PMAQ score by bonus size in the matched sample. PMAQ, National Programme for Improving Primary Care Access and Quality. (TIF)

S1 Data. Data used to generate Fig 1. (XLSX)

S2 Data. Data used to generate Fig 2. (XLSX)

S3 Data. Data used to generate Fig 3. (XLSX)

Acknowledgments

We thank Allan Nuno Alves de Sousa, Olivia Lucena, Davllyn Anjos, Ilano Barreto, and Wellington Carvalho for their valuable comments and insights throughout the project. We are grateful to the Ministry of Health of Brazil for sharing the data on the PMAQ score and for providing information on the design of PMAQ at the national level.

Author Contributions

Conceptualization: Nasser Fardousi, Everton Nunes da Silva, Josephine Borghi, Timothy Powell-Jackson.

Data curation: Nasser Fardousi, Everton Nunes da Silva, Roxanne Kovacs, Letícia Xander Russo, Timothy Powell-Jackson.

Formal analysis: Nasser Fardousi, Timothy Powell-Jackson.

Funding acquisition: Everton Nunes da Silva, Josephine Borghi, Timothy Powell-Jackson.

Investigation: Nasser Fardousi, Timothy Powell-Jackson.

Methodology: Nasser Fardousi, Timothy Powell-Jackson.

Project administration: Timothy Powell-Jackson.

Resources: Timothy Powell-Jackson.

Supervision: Timothy Powell-Jackson.

Visualization: Nasser Fardousi, Timothy Powell-Jackson.

Writing - original draft: Nasser Fardousi, Timothy Powell-Jackson.

Writing – review & editing: Nasser Fardousi, Everton Nunes da Silva, Roxanne Kovacs, Josephine Borghi, Jorge O. M. Barreto, Søren Rud Kristensen, Juliana Sampaio, Helena Eri Shimizu, Luciano B. Gomes, Letícia Xander Russo, Garibaldi D. Gurgel, Timothy Powell-Jackson.

References

- 1. World Health Organization. A vision for primary health care in the 21st century: towards universal health coverage and the Sustainable Development Goals. World Health Organization, 2018.
- Castro MC, Massuda A, Almeida G, Menezes-Filho NA, Andrade MV, de Souza Noronha KVM, et al. Brazil's unified health system: the first 30 years and prospects for the future. Lancet. 2019; 394 (10195):345–56. Epub 20190711. https://doi.org/10.1016/S0140-6736(19)31243-7 PMID: 31303318.
- Macinko J, Harris MJ. Brazil's family health strategy—delivering community-based primary care in a universal health system. N Engl J Med. 2015; 372(23):2177–81. <u>https://doi.org/10.1056/ NEJMp1501140 PMID: 26039598</u>
- Rasella D, Aquino R, Barreto ML. Reducing childhood mortality from diarrhea and lower respiratory tract infections in Brazil. Pediatrics. 2010; 126(3):e534–40. Epub 20100802. https://doi.org/10.1542/ peds.2009-3197 PMID: 20679307.
- Aquino R, de Oliveira NF, Barreto ML. Impact of the family health program on infant mortality in Brazilian municipalities. Am J Public Health. 2009; 99(1):87–93. Epub 20081113. https://doi.org/10.2105/AJPH. 2007.127480 PMID: 19008516; PubMed Central PMCID: PMC2636620.
- Rasella D, Harhay MO, Pamponet ML, Aquino R, Barreto ML. Impact of primary health care on mortality from heart and cerebrovascular diseases in Brazil: a nationwide analysis of longitudinal data. BMJ. 2014; 349:g4014. Epub 20140703. <u>https://doi.org/10.1136/bmj.g4014</u> PMID: <u>24994807</u>; PubMed Central PMCID: PMC4080829.
- Hone T, Rasella D, Barreto ML, Majeed A, Millett C. Association between expansion of primary healthcare and racial inequalities in mortality amenable to primary care in Brazil: a national longitudinal analysis. PLoS Med. 2017; 14(5):e1002306. https://doi.org/10.1371/journal.pmed.1002306 PMID: 28557989
- 8. Stopa SR, Malta DC, Monteiro CN, Szwarcwald CL, Goldbaum M, Cesar CLG. Use of and access to health services in Brazil, 2013 National Health Survey. Rev Saude Publica. 2017; 51(suppl 1):3.
- Facchini LA, Piccini RX, Tomasi E, Thume E, Teixeira VA, Silveira DS, et al. Evaluation of the effectiveness of Primary Health Care in South and Northeast Brazil: methodological contributions. Cad Saude Publica. 2008; 24 Suppl 1:S159–72. Epub 2008/07/29. <u>https://doi.org/10.1590/s0102-311x2008001300020</u> PMID: 18660900.
- Doran T, Maurer KA, Ryan AM. Impact of Provider Incentives on Quality and Value of Health Care. Annu Rev Public Health. 2017; 38:449–65. Epub 20161215. <u>https://doi.org/10.1146/annurev-publicalth-032315-021457</u> PMID: 27992731.
- Milstein R, Schreyoegg J. Pay for performance in the inpatient sector: A review of 34 P4P programs in 14 OECD countries. Health Policy. 2016; 120(10):1125–40. Epub 20160920. https://doi.org/10.1016/j. healthpol.2016.08.009 PMID: 27745916.
- Ogundeji YK, Bland JM, Sheldon TA. The effectiveness of payment for performance in health care: A meta-analysis and exploration of variation in outcomes. Health Policy. 2016; 120(10):1141–50. Epub 20160905. https://doi.org/10.1016/j.healthpol.2016.09.002 PMID: 27640342.
- Mendelson A, Kondo K, Damberg C, Low A, Motuapuaka M, Freeman M, et al. The Effects of Pay-for-Performance Programs on Health, Health Care Use, and Processes of Care: A Systematic Review. Ann Intern Med. 2017; 166(5):341–53. Epub 20170110. https://doi.org/10.7326/M16-1881 PMID: 28114600.
- Diaconu K, Falconer J, Verbel A, Fretheim A, Witter S. Paying for performance to improve the delivery of health interventions in low- and middle-income countries. Cochrane Database Syst Rev. 2021; 5(5): CD007899. Epub 20210505. https://doi.org/10.1002/14651858.CD007899.pub3 PMID: 33951190; PubMed Central PMCID: PMC8099148.
- Eijkenaar F. Key issues in the design of pay for performance programs. Eur J Health Econ. 2013; 14 (1):117–31. Epub 20110901. https://doi.org/10.1007/s10198-011-0347-6 PMID: 21882009; PubMed Central PMCID: PMC3535413.
- Kovacs RJ, Powell-Jackson T, Kristensen SR, Singh N, Borghi J. How are pay-for-performance schemes in healthcare designed in low- and middle-income countries? Typology and systematic literature review. BMC Health Serv Res. 2020; 20(1):291. Epub 20200407. https://doi.org/10.1186/s12913-020-05075-y PMID: 32264888; PubMed Central PMCID: PMC7137308.
- Ogundeji YK, Sheldon TA, Maynard A. A reporting framework for describing and a typology for categorizing and analyzing the designs of health care pay for performance schemes. BMC Health Serv Res. 2018; 18(1):686. Epub 20180904. https://doi.org/10.1186/s12913-018-3479-x PMID: <u>30180838</u>; PubMed Central PMCID: PMC6123918.
- Epstein AM. Will pay for performance improve quality of care? The answer is in the details. N Engl J Med. 2012. https://doi.org/10.1056/NEJMe1212133 PMID: 23134388

- Kristensen SR, Siciliani L, Sutton M. Optimal price-setting in pay for performance schemes in health care. J Econ Behav Organ. 2016; 123:57–77.
- Layton TJ, Ryan AM. Higher Incentive Payments in Medicare Advantage's Pay-for-Performance Program Did Not Improve Quality But Did Increase Plan Offerings. Health Serv Res. 2015; 50(6):1810–28. Epub 20151109. <u>https://doi.org/10.1111/1475-6773.12409</u> PMID: <u>26549194</u>; PubMed Central PMCID: PMC4693840.
- Navathe AS, Volpp KG, Caldarella KL, Bond A, Troxel AB, Zhu J, et al. Effect of Financial Bonus Size, Loss Aversion, and Increased Social Pressure on Physician Pay-for-Performance: A Randomized Clinical Trial and Cohort Study. JAMA Netw Open. 2019; 2(2):e187950. Epub 20190201. https://doi.org/10. 1001/jamanetworkopen.2018.7950 PMID: 30735234; PubMed Central PMCID: PMC6484616.
- Van Herck P, De Smedt D, Annemans L, Remmen R, Rosenthal MB, Sermeus W. Systematic review: Effects, design choices, and context of pay-for-performance in health care. BMC Health Serv Res. 2010; 10(1):247. Epub 20100823. https://doi.org/10.1186/1472-6963-10-247 PMID: 20731816; PubMed Central PMCID: PMC2936378.
- Singh NS, Kovacs RJ, Cassidy R, Kristensen SR, Borghi J, Brown GW. A realist review to assess for whom, under what conditions and how pay for performance programmes work in low-and middleincome countries. Soc Sci Med. 2020;113624. <u>https://doi.org/10.1016/j.socscimed.2020.113624</u> PMID: 33373774
- 24. Kandel E, Lazear EP. Peer pressure and partnerships. J Polit Economy. 1992; 100(4):801–17.
- Kristensen SR, Bech M, Lauridsen JT. Who to pay for performance? The choice of organisational level for hospital performance incentives. Eur J Health Econ. 2016; 17(4):435–42. Epub 20150410. https:// doi.org/10.1007/s10198-015-0690-0 PMID: 25860814.
- 26. Kovacs R, Barreto JOM, da Silva EN, Borghi J, Kristensen SR, Costa DRT, et al. Socioeconomic inequalities in the quality of primary care delivered by family health teams under Brazil's national payfor-performance programme. Lancet Glob Health. 2020.
- Brasil. Fundo Nacional de Saúde. Repasses Fundo a Fundo [09/03/2022]. Available from: https://painelms.saude.gov.br/extensions/Portal_FAF/Portal_FAF.html.
- 28. Secretaria de Atenção Primária à Saúde. Nota metodológica da certificação das equipes de atenção básica participantes do Programa de Melhoria do Acesso e da Qualidade na Atenção Básica. Brasília: Ministry of Health of Brazil; 2013.
- Secretaria de Atenção Primária à Saúde. Nota metodológica da certificação dos núcleos de apoio à saúde da família 2013–2014. Brasília: Ministry of Health of Brazil; 2014.
- 30. Secretaria de Atenção Primária à Saúde. Nota metodológica da certificação das equipes de atenção básica. Programa Nacional de Melhoria do Acesso e da Qualidade da Atenção Básica (PMAQ-AB)— Terceiro ciclo. Brasília: Ministry of Health of Brazil; 2018.
- **31.** Heckman JJ, Ichimura H, Todd PE. Matching as an econometric evaluation estimator: Evidence from evaluating a job training programme. Rev Econ Stud. 1997; 64(4):605–54.
- O'Neill S, Kreif N, Grieve R, Sutton M, Sekhon JS. Estimating causal effects: considering three alternatives to difference-in-differences estimation. Health Serv Outcome Res Methodol. 2016; 16(1):1–21. Epub 20160507. https://doi.org/10.1007/s10742-016-0146-8 PMID: 27340369; PubMed Central PMCID: PMC4869762.
- Sommers BD, Long SK, Baicker K. Changes in mortality after Massachusetts health care reform: a quasi-experimental study. Ann Intern Med. 2014; 160(9):585–93. <u>https://doi.org/10.7326/M13-2275</u> PMID: 24798521.
- McKenzie D. Revisiting the difference-in-differences parallel trends assumption: Part i pre-trend testing. World Bank Blogs. 2020.
- Daw JR, Hatfield LA. Matching and Regression to the Mean in Difference-in-Differences Analysis. Health Serv Res. 2018; 53(6):4138–56. Epub 20180629. https://doi.org/10.1111/1475-6773.12993 PMID: 29957834; PubMed Central PMCID: PMC6232412.
- Ryan AM. Well-Balanced or too Matchy-Matchy? The Controversy over Matching in Difference-in-Differences. Health Serv Res. 2018; 53(6):4106–10. Epub 20180725. https://doi.org/10.1111/1475-6773.13015 PMID: 30047128; PubMed Central PMCID: PMC6232433.
- Ho DE, Imai K, King G, Stuart EA. Matching as nonparametric preprocessing for reducing model dependence in parametric causal inference. Polit Anal. 2007; 15(3):199–236.
- Basinga P, Gertler PJ, Binagwaho A, Soucat AL, Sturdy J, Vermeersch CM. Effect on maternal and child health services in Rwanda of payment to primary health-care providers for performance: an impact evaluation. Lancet. 2011; 377(9775):1421–8. Epub 2011/04/26. https://doi.org/10.1016/S0140-6736 (11)60177-3 PMID: 21515164.

- **39.** Angrist JD, Pischke J-S. The credibility revolution in empirical economics: How better research design is taking the con out of econometrics. J Econ Perspect. 2010; 24(2):3–30.
- 40. Kahn-Lang A, Lang K. The promise and pitfalls of differences-in-differences: Reflections on 16 and pregnant and other applications. J Bus Econ Statist. 2020; 38(3):613–20.
- **41.** de Walque D, Gertler PJ, Bautista-Arredondo S, Kwan A, Vermeersch C, de Dieu Bizimana J, et al. Using provider performance incentives to increase HIV testing and counseling services in Rwanda. J Health Econ. 2015; 40:1–9. Epub 20141212. <u>https://doi.org/10.1016/j.jhealeco.2014.12.001</u> PMID: 25554976.
- Lannes L, Meessen B, Soucat A, Basinga P. Can performance-based financing help reaching the poor with maternal and child health services? The experience of rural Rwanda. Int J Health Plann Manag. 2016; 31(3):309–48. Epub 20150629. https://doi.org/10.1002/hpm.2297 PMID: 26122744.
- **43.** Friedman J, Qamruddin J, Chansa C, Das AK. Impact evaluation of Zambia's health results-based financing pilot project. Washington, DC: World Bank Group Working Paper No 120723. 2016.
- 44. de Walque D, Robyn PJ, Saidou H, Sorgho G, Steenland M. Looking into the performance-based financing black box: evidence from an impact evaluation in the health sector in Cameroon. Health Policy Plan. 2021; 36(6):835–47. https://doi.org/10.1093/heapol/czab002 PMID: 33963406.
- 45. Lagarde M, Burn S, Lawin L, Bello K, Makoutode P. Exploring the impact of performance-based financing on health workers' performance in Benin, 2015. Washington: World Bank, 2015.
- 46. Macinko J, Harris MJ, Rocha MG. Brazil's National Program for Improving Primary Care Access and Quality (PMAQ): Fulfilling the Potential of the World's Largest Payment for Performance System in Primary Care. J Ambul Care Manage. 2017; 40 Suppl 2 Supplement, The Brazilian National Program for Improving Primary Care Access and Quality (PMAQ):S4–S11. Epub 2017/03/03. https://doi.org/10. 1097/JAC.00000000000189 PMID: 28252498; PubMed Central PMCID: PMC5338882.
- Kristensen SR, Meacock R, Turner AJ, Boaden R, McDonald R, Roland M, et al. Long-term effect of hospital pay for performance on mortality in England. N Engl J Med. 2014; 371(6):540–8. https://doi. org/10.1056/NEJMoa1400962 PMID: 25099578.